
BLIP Operations - Sr-82 Production 

o  Safety - No operational events 

o  In production mode.  

o  Sr-82 and Rb-83 to be shipped out this week. 

o  Maintenance – thanks to all involved  

o  New Variable Speed Drives – only one replaced this week (F1). 

o  New Chiller installed commissioning complete. 

o  BLIP beam on 

o  200 MeV tuned on 19 Mar 2014 

o  Nick Simos LHC targets in Box 1 and Sr-82 targets in Box 2 (in beam on 

20 Mar). 

o  Anticipated 110 uA – but yesterday at 100 uA – will need to revise LHC 

research irradiations time scale and beam energy change.   
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Days of operation. 

Current	
  delivered	
  to	
  BLIP	
  18	
  Dec	
  –	
  24	
  Mar	
  2014	
  	
  
Significant drop in 
current 100 uA. 
 
Running below 
expectations. 
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  loaded	
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We	
  will	
  need	
  to	
  pull	
  LHC	
  target	
  
sooner	
  due	
  to	
  drop	
  in	
  current.	
  	
  	
  



Total Integrated Current at BLIP vs Days of Operation 

Days of operation. 
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